

Citation: Bisanzio D, Dzul-Manzanilla F, Gomez-Dantés H, Pavia-Ruz N, Hladish TJ, Lenhart A, et al. (2018) Spatio-temporal coherence of dengue, chikungunya and Zika outbreaks in Merida, Mexico. PLoS Negl Trop Dis 12(3): e0006298. https://doi. org/10.1371/journal.pntd.0006298

Editor: Nikos Vasilakis, University of Texas Medical Branch, UNITED STATES

Received: December 5, 2017

Accepted: February 4, 2018

Published: March 15, 2018

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: Data are available at Open Science Framework (OSF): Bisanzio, D. (2017). DENV, CHIKV, ZIKV in Merida 2008–2016; project number: FNKXD; URL:osf.io/fnkxd/.

Funding: Research funding was provided by the National Science Foundation (DEB/EEID award: 1640698), the Office of Infectious Disease, Bureau for Global Health, U.S. Agency for International Development, under the terms of an Interagency Agreement with the US Centers for Disease Control and Prevention (CDC: OADS BAA 2016-N-17844),

RESEARCH ARTICLE

Spatio-temporal coherence of dengue, chikungunya and Zika outbreaks in Merida, Mexico

Donal Bisanzio^{1,2}, Felipe Dzul-Manzanilla³, Hector Gomez-Dantés⁴, Norma Pavia-Ruz⁵, Thomas J. Hladish^{6,7}, Audrey Lenhart⁸, Jorge Palacio-Vargas⁹, Jesus F. González Roldan³, Fabian Correa-Morales³, Gustavo Sánchez-Tejeda³, Pablo Kuri Morales¹⁰, Pablo Manrique-Saide¹¹, Ira M. Longini^{7,12,13}, M. Elizabeth Halloran^{13,14,15}, Gonzalo M. Vazquez-Prokopec¹⁶*

1 RTI International, Washington, DC, United States of America, 2 Centre for Tropical Diseases, Sacro Cuore-Don Calabria Hospital, Negrar, Verona, Italy, 3 Centro Nacional de Programas Preventivos y Control de Enfermedades (CENAPRECE) Secretaría de Salud Mexico, Mexico City, Mexico, 4 Health Systems Research Center, National Institute of Public Health, Cuernavaca, Mexico, 5 Centro de Investigaciones Regionales Hideyo Noguchi, Universidad Autonoma de Yucatan, Merida, Mexico, 6 Department of Biology, University of Florida, Gainesville, FL, United States of America, 7 Emerging Pathogens Institute, University of Florida, Gainesville, FL, United States of America, 8 Centers for Disease Control and Prevention, Atlanta, GA, United States of America, 9 Secretaria de Salud de Yucatan. Merida, Mexico, 10 Subsecretaría de Prevención y Promoción de la Salud, Mexico City, Mexico, 11 Unidad Colaborativa de Bioensayos Entomológicos, Campus de Ciencias Biológicas y Agropecuarias, Universidad Autónoma de Yucatán, Merida, Mexico, 12 Department of Biostatistics, University of Florida, Gainesville, FL, United States of America, 13 Center for Inference and Dynamics of Infectious Diseases, Seattle, WA, United States of America, 14 Vaccine and Infectious Disease Division, Fred Hutchinson Cancer Research Center, Seattle, WA, United States of America, 15 Department of Biostatistics, University of Washington, Seattle, WA, United States of America, 16 Department of Environmental Sciences, Emory University, Atlanta, GA, United States of America

* gmvazqu@emory.edu

Abstract

Response to Zika virus (ZIKV) invasion in Brazil lagged a year from its estimated February 2014 introduction, and was triggered by the occurrence of severe congenital malformations. Dengue (DENV) and chikungunya (CHIKV) invasions tend to show similar response lags. We analyzed geo-coded symptomatic case reports from the city of Merida, Mexico, with the goal of assessing the utility of historical DENV data to infer CHIKV and ZIKV introduction and propagation. About 42% of the 40,028 DENV cases reported during 2008–2015 clustered in 27% of the city, and these clustering areas were where the first CHIKV and ZIKV cases were reported in 2015 and 2016, respectively. Furthermore, the three viruses had significant agreement in their spatio-temporal distribution (Kendall W>0.63; p<0.01). Longitudinal DENV data generated patterns indicative of the resulting introduction and transmission patterns of CHIKV and ZIKV, leading to important insights for the surveillance and targeted control to emerging *Aedes*-borne viruses.

and by partial support from the National Institutes of Health through the MIDAS research network (U54 GM111274) and the Canadian Institutes of Health Research (CIHR) and IDRC (Preventing Zika disease with novel vector control approaches, Project 108412). 'SANOFI Pasteur financed the cohort study (DNG25) for which HGD and NPR were PIs. The opinions expressed herein are those of the author(s) and do not necessarily reflect the views of the U.S. Agency for International Development or the Centers for Disease Control and Prevention. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Over the past decades, *Aedes*-borne viruses (dengue, chikungunya, Zika) have become a major source of morbidity within urban areas. Worldwide, public health response to these viruses is reactive to the occurrence of symptomatic cases (a small proportion of all infections). Here we used geocoded passive surveillance data to determine if areas of historically persistent dengue transmission fuel the introduction and propagation of other *Aedes*-borne viruses. This article provides quantitative evidence of the strong spatio-temporal overlap that occurs between dengue, chikungunya and Zika, all transmitted by *Aedes* aegypti mosquitoes in the city. Additionally, it emphasizes the value of analyzing long-term geo-coded passive surveillance information to help identify areas for prioritizing surveillance and control. Findings from this article open a window for considering historical DENV data to make predictions of likely sources of invasion for other emerging *Aedes*-borne viruses, as well as to the consideration of spatially-targeted approaches for delivery of vector control and surveillance. Arbovirus control in complex urban environments can greatly benefit from exploiting existing spatial information for better delivery of interventions.

Introduction

The unprecedented global emergence of infectious disease threats has demonstrated how illprepared the world is to predict, rapidly respond to, and contain pandemics [1–3]. Zika virus (ZIKV), an emerging arthropod-borne flavivirus, has become a global public health threat since its emergence outside Africa in 2007 [4,5]. A large outbreak in French Polynesia in 2013–2014 showed that an otherwise clinically mild illness (common symptoms include mild fever, rash, arthralgia, and conjunctivitis) could also be responsible for severe neurological complications in adults (e.g., Guillain-Barre syndrome) [6,7]. Once active ZIKV transmission was recognized in Brazil in 2015–2016, the spectrum of disease severity expanded to include neurodegenerative complications in newborns (e.g., microcephaly) as well as other developmental sequelae grouped within the diagnostic term "congenital Zika syndrome" [4,8].

The World Health Organization declaration of Zika as a public health emergency of international concern on February 1, 2016, led to a global coordinated effort to improve vaccine development and scale-up vector control actions against the primary ZIKV vector in the Americas, *Aedes aegypti*. Failure in previous years to contain the transmission of chikungunya (CHIKV) and dengue (DENV) viruses, also primarily transmitted by *Ae. aegypti*, demonstrated the challenge in scaling-up vector control tools and interventions such as ultra-low volume (ULV) insecticide applications, larviciding and community mobilization, all with limited efficacy in preventing disease [9]. Although most DENV endemic countries in the Americas anticipated the arrival of ZIKV, they still performed only reactive interventions once symptomatic cases (assumed to be ~20% of all ZIKV infections, [4]) were detected. This led to significant delays in response and limited impact on virus containment, as seen in the rapid geographic spread of epidemics [5]. While some vaccine candidates may soon enter into Phase IIb/III trials [10], vector control and community education still remain the main preventive approaches to minimize the severe and devastating health outcomes associated with ZIKV infection.

DENV, CHIKV and ZIKV are all primarily transmitted by *Ae. aegypti* mosquitoes, leading to the early assumption that ZIKV would follow the path of the other two viruses in the Americas [11]. While this hypothesis is supported at regional scales, with the three viruses following

similar transmission seasonality and regional circulation (e.g., [12]), it is unknown how their transmission pattern could differ within urban centers. DENV tends to show strong heterogeneity within cities, with some neighborhoods reporting higher sero-prevalence or transmission than others (e.g., [13–17]). Daytime human movement and aggregation patterns can influence exposure and are drivers of both human-*Ae. aegypti* contacts and rapid virus propagation, affecting DENV, CHIKV and ZIKV transmission [18–21]. Given the shared vector and ecology between the three viruses, we conducted spatio-temporal analyses on historical (2008–2015) DENV passive surveillance data and recent (2015–2016) CHIKV and ZIKV virus invasion events within the city of Merida, Mexico, and used the results to answer the following questions: Are DENV transmission hot-spots within cities also likely to be CHIKV and ZIKV transmission hot-spots? Do areas of historically persistent DENV transmission fuel the introduction and propagation of the other two viruses?

Materials and methods

Study site

The Yucatan Peninsula of Mexico is endemic for all dengue serotypes, with DENV1 and DENV2 dominating during 2008–2015 and DENV4 invading in 2013. Over the same period, 40,028 probable DENV cases have been reported from the city of Merida (S1 Text), the most important urban center (population ~1 million) and Yucatan State's capital, and major contributor to the burden of virus transmission in the region [22]. Merida is located in a subtropical environment with mean temperatures ranging from 29°C in December to 34°C in July. The rainy season occurs from May to October and overlaps with the peak dengue transmission season between July and November, although cases occur year-round [23]. Dengue virus is widely distributed throughout the Yucatan peninsula, and the vector control strategies used by local authorities include ultra-low volume (ULV) spraying with the organophosphate insecticides chlorpyrifos and malathion and indoor space spraying with pyrethroids (deltamethrin) and organophosphates (malathion).

Data sources and analysis

Recent operational innovations in the surveillance and control of *Aedes*-borne viruses in Mexico led to the development of a comprehensive online system for the capture, visualization and analysis of entomologic and epidemiologic data [24]. The National System of Epidemiological Surveillance of the Mexican Ministry of Health (operational since 2008) includes a nationwide, web-based, geographic information system where disease-specific reported DENV, CHIKV and ZIKV cases are automatically geocoded and mapped at the geographic scale of households [24]. The following information was obtained for the city of Merida: probable and laboratory confirmed cases geocoded to the household level and with information of age, epidemiologic week of the onset of symptoms for every case; disease infection status (DENV, CHIKV, ZIKV) for each case; DENV serotype (for a subset of laboratory-confirmed cases). Demographic information was obtained for the city of Merida (2010 census) from Mexico's National Census and aggregated at the level of census tracts, called Area Geoestadistica Basica (AGEB). Tracts aggregate up to 50 city blocks (average area, 0.5 km²; Standard Deviation, 0.3 km²), and Merida has 540 tracts.

Epidemiologic information was aggregated at the spatial scale of census tracts and the temporal scale of weeks, with separate datasets generated for each virus. The weekly count of cases was normalized, to allow for comparisons between years and viruses, by calculating, for each virus, weekly z-scores per tract as follows:

$$z_i = \frac{N_i - \mu}{\sigma} \tag{1}$$

where N_i is the number of cases in tract *i*, μ the mean case count for the city, and σ the standard deviation of case counts for the city. For a given week, the obtained z-score was rescaled dividing it by census tract's maximum value of case counts.

The weekly trend in DENV cases was analyzed to separate, during the years 2008–2015, the epidemic from non-epidemic transmission periods of the time-series using change point detection methods [25]. A G^* local spatial clustering test [26] was applied to annual normalized case counts separately for epidemic and non-epidemic periods (for DENV) and by year for CHIKV and ZIKV. The significance of the computed G^* was evaluated by comparing observed values to the null hypothesis of random case distribution by randomly re-assigning the tract labels to the case counts. This was performed using 100,000 Monte Carlo randomizations. Tracts with statistically significant (p<0.05, calculated adopting Bonferroni correction) high z-scores, defined as hot-spots, were mapped for each virus and epidemic period (DENV) or year (CHIKV, ZIKV). For DENV, G^* output was aggregated to estimate a count of time-periods that a tract was identified as a hot-spot.

A Kendall *W* test evaluated the spatio-temporal overlap between DENV, CHIKV and ZIKV z-values by tract. *W* measures concordance between two datasets, ranging from +1 (complete agreement) to -1(no agreement) and was calculated for all combinations of virus pairs (DENV-CHIKV, DENV-ZIKV, CHIKV-ZIKV). A second approach to evaluate spatio-temporal overlap between viruses consisted in overlaying the persistence of DENV hot-spots per tract with the hot-spots identified for CHIKV (2015, 2016) and for ZIKV (2016).

Data from a longitudinal DENV cohort study following 1,666 residents aged 1-60 from Merida was utilized to contrast DENV seroprevalence of those living in areas identified as hotspots versus those living outside. Briefly, 17 elementary schools distributed across Merida in areas thought to differ in DENV incidence (low incidence in the north and high incidence in the south) were chosen as the hubs for participant enrollment and follow-up. From the school student list, 464 children were selected (up to 50 per grade per school) for the serological follow-up, which also included all family members living in the same residence and who consented to the study. Annual blood draws were performed in 2015 and positivity to any DENV serotype was assessed by Yucatan State Laboratories. Prior exposure to dengue and age-specific serostatus were determined using Panbio IgG indirect ELISA. Standard cut-off points were used for defining positive (\geq 12 Panbio units) from negative samples (<9 Panbio units) and indeterminate for those in between. From the people enrolled in the cohort, we selected those whose residence was found inside the areas identified as DENV hot-spots and randomly selected a similar number of individuals enrolled in the cohort living outside the area. As our retrospective analysis covered the period 2008-2015, we focused our analysis on children aged 8 years or younger. A mixed-effects logistic regression model quantified the odds ratio for the following fixed effects: DENV cluster (i.e., living outside or inside the 2008-2015 DENV hotspot areas) and age. A random intercept was included at the level of the household to account for the occurrence of nested samples within each house. Models were run with the R package lme4 [27].

Ethics statement

Protocols for processing and analyzing data were approved by Emory University's ethics committee under protocol ID: IRB00088659. The protocol was also approved by the Ethics and Research Committee from the O'Horan General Hospital from the state Ministry of Health, Register No. CEI-0-34-1-14. All cases were anonymized before being analyzed.

Results

Spatio-temporal dynamics of DENV transmission

A total of 40,028 clinically apparent DENV cases were reported to the public health system of Merida during 2008–2015, 94.5% (37,894) occurring in eight epidemic periods lasting on average 34 weeks (range: 24–44) (Fig A in S1 Text). Approximately 30% of all cases were confirmed by laboratory assays (IgM and PCR). During this period, three DENV serotypes circulated in Merida; DENV1 predominated in most years, DENV2 invaded in 2009 and DENV4 in 2013 (Fig B in S1 Text). About 40% of reported cases occurred in <15 year olds, with the proportion of cases by age-class not differing significantly between years (Fig C in S1 Text) (Fisher's exact test, p < 0.05).

The weekly spatio-temporal progression of standardized DENV case counts within the city was heterogeneous across census tracts, showing a complex pattern of emergence and persistence over the 8-year period (Fig 1A and 1B, Fig D in S1 Text). When summarized over the entire period, standardized case counts aggregated in nearby tracts (Fig 1C) suggested the occurrence of strong autocorrelation. The Getis-Ord *G*^{*} statistic, calculated separately for epidemic and non-epidemic periods (Fig E and Fig F in S1 Text), identified tracts within Merida with persistent DENV transmission up to 6 years (epidemic, Fig 1D) and 4 years (non-epidemic, Fig 1E) concentrated in the center of the city. Altogether, the hot-spot areas contained 41.9% (16,773) of all reported symptomatic cases within 27.8% of the city's 540 census tracts.

Serological status for DENV infection in 505 children aged 8 or younger belonging to a longitudinal cohort study was compared between those residing inside and outside the hot-spot areas (244 children inside and 261 children outside the area) (Fig 2A). Living inside the hotspot area was associated with a significantly higher infection probability than living outside (odds ratio, 1.71 [95%CI, 1.08–2.20]; p < 0.05), after adjusting for the age of the child and the occurrence of nested cases within households (Table 1). Model-predicted DENV infection probabilities were significantly higher in children aged 6–8 years living inside a hot-spot area, compared to children the same age living outside (Fig 2B).

Association with CHIKV and ZIKV transmission

In 2015, the first CHIKV case was reported in a tract within the persistent DENV clustering area (Fig G in <u>S1 Text</u>). From this tract, the virus propagated rapidly throughout the city (counting 1,101 probable cases) (Fig G in <u>S1 Text</u>), leading to difficulties in containing both DENV and CHIKV. ZIKV was first reported in April 2016, also inside the persistent DENV clustering area (Fig G in <u>S1 Text</u>). The 2016 ZIKV outbreak included 2,273 reported cases (of which ~50% were in <15 year olds, Fig H in <u>S1 Text</u>). No Zika-related microcephaly cases were yet detected in newborns to mothers infected during this first wave of ZIKV invasion. The tracts where the first 10 CHIKV and ZIKV cases were reported overlapped 100% and 75%, respectively, with the DENV persistent clustering area (Fig G in <u>S1 Text</u>).

The DENV, CHIKV and ZIKV case counts by tract showed a significantly (p<0.01) high coherence, with Kendall's W values of 0.75 for DENV-ZIKV comparisons, 0.72 for DENV-CHIKV and 0.63 for CHIKV-ZIKV. The high similarity among DENV, CHIKV, and ZIKV are also evident when comparing distribution of standardized case counts using quantile-quantile (Q-Q) plots (Fig 3A). A 35.6% (392) of all reported CHIKV cases and 66.9% (1,509) of all ZIKV cases were found within the area of persistent DENV clustering. When comparing the location and extent of hot-spots, 9 (64.3%) CHIKV clusters for 2015 and 2016 fully



Fig 1. Spatio-temporal distribution of standardized dengue virus (DENV) cases in Merida, Mexico, during 2008–2015. (A) Weekly case counts are shown as bar plot (above) and by census tract (each row is a tract and rows are arranged by geographic proximity, with near units being next to each other, as shown in the color bar located on the right margin which matches the color map in (B)). Right panel shows total count of dengue cases per census unit during 2008–2015, which is mapped in (C). Panel (D)

shows the frequency of outbreak periods in which each census unit was found to be a statistically significant hot-spot of DENV transmission. Panel (E) shows the same information for non-outbreak periods. Source of the census tract boundaries was Instituto Nacional de Estadística y Geografía (INEGI), 2010.

https://doi.org/10.1371/journal.pntd.0006298.g001

overlapped with the 2008–2015 DENV clustering, and 3 (14.3%) clusters of high ZIKV standardized case counts overlapped with the DENV clustering area (Fig 3A).

Discussion

Combining powerful analytic methods with detailed retrospective data provides evidence of the significant within-city spatio-temporal overlap between three *Aedes*-borne viruses. While forecasting the exact timing and introduction points of CHIKV and ZIKV within a city would be bound with significant uncertainty [28], historical spatio-temporal DENV surveillance data can be analyzed to identify the extent of high transmission areas that would serve as important transmission foci for novel *Aedes*-borne viruses.

Surveillance data has characterized the temporal pattern of CHIKV and ZIKV invasion in areas with active DENV transmission in great detail (e.g., [6,12]). Such descriptive studies provide evidence of high temporal overlap between the three viruses, with transmission peaks of CHIKV and ZIKV occurring during the same time periods as DENV peaks. In Merida, the three viruses not only overlapped in time, but they also exhibited strong spatio-temporal coherence in their distribution. The first CHIKV and ZIKV cases occurred within the DENV clustering area with a strong match between pathogens. It is important to note the partial



Fig 2. Validation of DENV spatial clustering. (A) location of the residence of 505 children aged 8 or younger and who provided a blood sample in 2015, stratified by those residing inside (black) and outside (red) the 2008–2015 DENV hot-spot area. (B) Average DENV infection probability, estimated from a mixed-effects logistic regression model, for the children living inside (black solid line) or outside (red dotted line) the 2008–2015 DENV hot-spot area. Error bars indicate the standard error of the mean. Details of the model are shown in Table 1. Source of the census tract boundaries was INEGI, 2010.

https://doi.org/10.1371/journal.pntd.0006298.g002

PLOS | NEGLECTED TROPICAL DISEASES

Odds Ratio Estimate 2.5% 97.5% P-value Z-value Parameter DENV Cluster* 1.710 1.078 2.861 2.199 0.0279 1.739 1.471 2.133 5.918 < 0.001 Age** 0.010 0.002 0.036 -6.596 < 0.001 Intercept

Table 1. Odds ratio estimated from a mixed-effects logistic regression model evaluating the association between DENV seroprevalence in cohort participants aged 8 or younger and their location of residence (2008–2015 DENV clustering area).

* Residence of participant found outside the 2008–2015 DENV clustering area was used as reference.

** Age at the time of providing a blood sample.

https://doi.org/10.1371/journal.pntd.0006298.t001

mismatch between ZIKV and DENV hot-spots. While stochasticity in the initial virus introduction point may contribute to this finding, it is worth mentioning that in 2016 the Yucatan Ministry of Health intensified vector control interventions (i.e., indoor space spraying, ULV spraying and the use of larvicide) in the same areas predicted as hot-spots (based on our preliminary maps) as a way to potentially slow-down the spread of the virus. Whether this enhanced control contributed to the reduced frequency of ZIKV clustering in DENV hotspots (even in the early presence of cases in the area) will require further study.



Fig 3. Spatial pattern of arbovirus hot-spots in Merida. (A) Hot-spots of CHIKV (2015–2016) and ZIKV (2016) standardized case counts overlapped with persistent DENV clustering area (gray polygons represent persistent DENV clustering areas and red polygons represent statistically significant yearly hot-spots for CHIKV and ZIKV). (B) Q-Q plots comparing distribution of standardized case counts of DENV, CHIKV, and ZIKV. Source of the census tract boundaries was INEGI, 2010.

https://doi.org/10.1371/journal.pntd.0006298.g003

Mathematical models predict that, for pathogens that show strong spatial or social heterogeneity, targeted interventions would represent a more effective means of disease mitigation as compared to blanket control [29,30]. Focusing efforts in high transmission areas has the additional benefits of increased efficiency by better allocating limited personnel and resources [31]. Recognizing areas of persistent DENV, CHIKV or ZIKV transmission can enable the preemptive deployment of vector control before transmission is apparent [32]. The possible benefits of such pro-active, targeted control may actually be experienced on a larger spatial scale, as interventions may additionally protect individuals who visit the hot-spot areas but do not reside in them [33]. Thus, in settings where Aedes-borne diseases are seasonal, targeting vector control to traditional hot-spot areas in an anticipatory fashion during the period of low transmission can lead to efficiencies in intervention coverage (when personnel workload is at its lowest) and theoretically provide the highest intervention effectiveness (~60-90% of cases prevented) [34]. Alternatively, vector habitat modification strategies such as housing improvement [35,36] may be more cost-effective and justifiable if undertaken within hot-spots. Before targeted control can be broadly recommended, our findings will have to be validated in other urban centers (ideally, of differing size and level of DENV endemicity), and the epidemiological impact of targeting hot-spots will need to be empirically evaluated and weighed against the value of reactive interventions. Additionally, any targeted intervention will have to be monitored for potential future changes in epidemiological trends, like the emergence of new hotspots in the city due to shifts vector abundance or socio-environmental conditions.

Several issues limit the quality and power of passive surveillance data and the potential for their use to forecast DENV transmission [28,37]. The data do not capture the disproportionately large number of infections that occur sub-clinically or that are mild enough not to result in a visit to a doctor. In many areas, DENV may be misdiagnosed as other febrile illnesses (e.g., influenza, leptospirosis [38]), adding uncertainty to observed trends. Additionally, given Ae. aegypti is a day-biting mosquito, mapping the place of residence may not correspond to the location where infection occurred [18,39]. Finally, in cities where Ae. aegypti and Ae. albopictus co-inhabit, spatial patterns of arbovirus transmission may be potentially influenced by their apparent disjoint distribution within urban areas (e.g., [40]). In Merida, where Ae. aegypti is the only vector and Ae. albopictus is not present, aggregating case data to relatively small (up to 50 city blocks) geographic units and analyzing standardized data over multiple years provided a strong signature of the spatial pattern of DENV circulation. About 27.8% (150) of the city districts accounted for 41.9% of cases, and such transmission heterogeneity was independently captured epidemiologically in the age-dependent sero-prevalence of DENV infection in children aged 8 years or younger. The ability of spatial clustering tests performed on passive surveillance data to capture strong trends in virus sero-prevalence provide a prospect for the development of city-wide risk maps, particularly if analyses involve multiple years and several virus introduction events.

Several spatial epidemiology studies have determined the set of census-derived environmental and socio-economic variables associated with DENV hot-spots. For instance, in Machala (Ecuador) hot-spots of dengue incidence at the level of census tracts was associated with older age and female gender of the head of the household, greater access to piped water in the home, poor housing condition, and less distance to the central hospital ([41]). In an endemic city in Venezuela, DENV hot-spots at the block level were associated with living in crowded conditions, having an occupation of domestic worker/housewife and not using certain preventive measures against mosquitoes ([42]). The contrast between these and other spatial epidemiology studies point to an important issue related to the finding of a generalized explanation to the elevated risk of DENV infection in hot-spots. Census variables vary from country to country, studies use data at different levels of aggregation, many census variables are likely auto-correlated, and there is an almost absence of entomological information. Thus, our study excluded imputing hot-spots to specific census variables because we will be performing a comprehensive environmental and anthropological assessment of the root drivers of the epidemiological trends observed in Merida.

We acknowledge several limitations to our study. While we used residential addresses in our analyses, we are aware that this excludes the potential for infection in areas other than the home [18,21]. Thus, our study was not able to explicitly account for movement and exposure in locations other than the residence, which may have obscured both the passive surveillance and the serological data. Future human mobility studies in Merida will help elucidate the role of hot-spots as foci of infection from both their residents and visitors from other parts of the city. The lack of Zika-related microcephaly cases in Merida at the point of analysis prevented evaluating any possible overlap between them and the historical DENV hot-spot areas. Given the possibility for immune interaction between viruses ([43]), evaluating if areas of intense DENV transmission lead to higher risk of microcephaly in newborns can lead to important insights for disease surveillance and control. While laboratory diagnosis of all three viruses is performed routinely in Mexico, current policy dictates that testing should be reduced once outbreaks are declared. This prompted us to include in our analyses clinically diagnosed (i.e., without laboratory confirmation) cases in addition to those with a laboratory confirmation.

While vaccines are seen as an ultimate disease mitigation strategy, their development can lag years after pathogen emergence [44], leaving case management and outbreak containment as the first line of defense. Thus, public health efforts face a challenging prospect: identifying vulnerable populations or likely transmission hot-spots as an approach to pro-actively mitigate emerging pathogen introduction and propagation. Recognizing the critical need to improve surveillance to prevent and rapidly contain pathogen transmission, the World Health Organization (WHO) is developing a global coordination mechanism for the containment of emerging pathogens [3]. Crucial to such effort is the integration of surveillance data and robust analytical methods to anticipate and better respond to disease threats [45,46]. Leveraging historical DENV, CHIKV and ZIKV data can help endemic countries improve control and surveillance programs by acknowledging and accounting for inherent spatial variability in risk within urban areas. Downscaling existing global risk maps (e.g., [47,48]) to capture within-city transmission risk is an important next step in disease mapping, as this can provide policy-makers with a more precise tool for intervention planning and strategic deployment at an operational scale.

Supporting information

S1 Text. Supporting figures showing: A) weekly count of DENV symptomatic cases reported to the public health system; B) relative frequency of isolated DENV serotypes; C) age-structure of reported DENV cases; D) standardized DENV case counts by year and census tract; E and F) results from the Getis-Ord G^* statistic test; G) time series of weekly case counts for CHIKV (A) and ZIKV (B) in Merida; H) age-structure of reported CHIKV and ZIKV. (PDF)

Acknowledgments

We acknowledge the DENTARGET network (<u>http://dentarget.finlay.edu.cu/</u>) for insightful discussion about DENV spatial distribution and dengue clustering.

Author Contributions

Conceptualization: Norma Pavia-Ruz, Pablo Manrique-Saide, Gonzalo M. Vazquez-Prokopec. Data curation: Norma Pavia-Ruz.

Formal analysis: Donal Bisanzio, Gonzalo M. Vazquez-Prokopec.

- Funding acquisition: Hector Gomez-Dantés, Pablo Manrique-Saide, Gonzalo M. Vazquez-Prokopec.
- **Investigation:** Felipe Dzul-Manzanilla, Hector Gomez-Dantés, Thomas J. Hladish, Gonzalo M. Vazquez-Prokopec.
- Methodology: Donal Bisanzio, Norma Pavia-Ruz, Thomas J. Hladish, Pablo Manrique-Saide, Ira M. Longini, M. Elizabeth Halloran, Gonzalo M. Vazquez-Prokopec.
- Project administration: Pablo Manrique-Saide, Gonzalo M. Vazquez-Prokopec.
- **Resources:** Jorge Palacio-Vargas, Jesus F. González Roldan, Fabian Correa-Morales, Gustavo Sánchez-Tejeda, Pablo Kuri Morales.

Software: Donal Bisanzio.

- Supervision: Norma Pavia-Ruz, Audrey Lenhart, Pablo Manrique-Saide, Gonzalo M. Vazquez-Prokopec.
- Validation: Donal Bisanzio, Gonzalo M. Vazquez-Prokopec.
- Visualization: Donal Bisanzio, Gonzalo M. Vazquez-Prokopec.
- Writing original draft: Donal Bisanzio, Gonzalo M. Vazquez-Prokopec.
- Writing review & editing: Thomas J. Hladish, Audrey Lenhart, Jorge Palacio-Vargas, Pablo Manrique-Saide, Ira M. Longini, M. Elizabeth Halloran.

References

- Jones KE, Patel NG, Levy MA, Storeygard A, Balk D, et al. (2008) Global trends in emerging infectious diseases. Nature 451: 990–993. https://doi.org/10.1038/nature06536 PMID: 18288193
- Morens DM, Fauci AS (2013) Emerging infectious diseases: threats to human health and global stability. PLoS Pathog 9: e1003467. https://doi.org/10.1371/journal.ppat.1003467 PMID: 23853589
- 3. Kieny MP, Salama P WHO R&D Blueprint: a global coordination mechanism for R&D preparedness. The Lancet 389: 2469–2470.
- Musso D, Gubler DJ (2016) Zika Virus. Clin Microbiol Rev 29: 487–524. https://doi.org/10.1128/CMR. 00072-15 PMID: 27029595
- Zhang Q, Sun K, Chinazzi M, Pastore YPA, Dean NE, et al. (2017) Spread of Zika virus in the Americas. Proc Natl Acad Sci U S A 114: E4334–E4343. https://doi.org/10.1073/pnas.1620161114 PMID: 28442561
- 6. Roth A, Mercier A, Lepers C, Hoy D, Duituturaga S, et al. (2014) Concurrent outbreaks of dengue, chikungunya and Zika virus infections—an unprecedented epidemic wave of mosquito-borne viruses in the Pacific 2012–2014. Euro Surveill 19.
- Cao-Lormeau VM, Blake A, Mons S, Lastere S, Roche C, et al. (2016) Guillain-Barre Syndrome outbreak associated with Zika virus infection in French Polynesia: a case-control study. Lancet 387: 1531– 1539. https://doi.org/10.1016/S0140-6736(16)00562-6 PMID: 26948433
- de Araujo TV, Rodrigues LC, de Alencar Ximenes RA, de Barros Miranda-Filho D, Montarroyos UR, et al. (2016) Association between Zika virus infection and microcephaly in Brazil, January to May, 2016: preliminary report of a case-control study. Lancet Infect Dis 16: 1356–1363. https://doi.org/10.1016/ S1473-3099(16)30318-8 PMID: 27641777
- Bowman LR, Donegan S, McCall PJ (2016) Is Dengue Vector Control Deficient in Effectiveness or Evidence?: Systematic Review and Meta-analysis. PLoS Negl Trop Dis 10: e0004551. <u>https://doi.org/10.1371/journal.pntd.0004551</u> PMID: 26986468
- Marston HD, Lurie N, Borio LL, Fauci AS (2016) Considerations for Developing a Zika Virus Vaccine. N Engl J Med 375: 1209–1212. https://doi.org/10.1056/NEJMp1607762 PMID: 27682031

- Musso D, Cao-Lormeau VM, Gubler DJ (2015) Zika virus: following the path of dengue and chikungunya? Lancet 386: 243–244. https://doi.org/10.1016/S0140-6736(15)61273-9 PMID: 26194519
- Cardoso CW, Paploski IAD, Kikuti M, Rodrigues MS, Silva MMO, et al. (2015) Outbreak of Exanthematous Illness Associated with Zika, Chikungunya, and Dengue Viruses, Salvador, Brazil. Emerging Infectious Diseases 21: 2274–2276. https://doi.org/10.3201/eid2112.151167 PMID: 26584464
- Vazquez-Prokopec GM, Kitron U, Montgomery B, Horne P, Ritchie SA (2010) Quantifying the spatial dimension of dengue virus epidemic spread within a tropical urban environment. PLoS Negl Trop Dis 4: e920. https://doi.org/10.1371/journal.pntd.0000920 PMID: 21200419
- Liebman KA, Stoddard ST, Morrison AC, Rocha C, Minnick S, et al. (2012) Spatial Dimensions of Dengue Virus Transmission across Interepidemic and Epidemic Periods in Iquitos, Peru (1999–2003). PLoS Neglected Tropical Diseases 6: e1472. https://doi.org/10.1371/journal.pntd.0001472 PMID: 22363822
- Morrison AC, Minnick SL, Rocha C, Forshey BM, Stoddard ST, et al. (2010) Epidemiology of Dengue Virus in Iquitos, Peru 1999 to 2005: Interepidemic and Epidemic Patterns of Transmission. PLoS Neglected Tropical Diseases 4: e670. https://doi.org/10.1371/journal.pntd.0000670 PMID: 20454609
- Pepin KM, Leach CB, Marques-Toledo C, Laass KH, Paixao KS, et al. (2015) Utility of mosquito surveillance data for spatial prioritization of vector control against dengue viruses in three Brazilian cities. Parasites & Vectors 8: 98.
- Salje H, Lessler J, Endy TP, Curriero FC, Gibbons RV, et al. (2012) Revealing the microscale spatial signature of dengue transmission and immunity in an urban population. Proceedings of the National Academy of Sciences of the United States of America 109: 9535–9538. https://doi.org/10.1073/pnas. 1120621109 PMID: 22645364
- Stoddard ST, Forshey BM, Morrison AC, Paz-Soldan VA, Vazquez-Prokopec GM, et al. (2013) Houseto-house human movement drives dengue virus transmission. Proc Natl Acad Sci U S A 110: 994–999. https://doi.org/10.1073/pnas.1213349110 PMID: 23277539
- Grubaugh ND, Ladner JT, Kraemer MUG, Dudas G, Tan AL, et al. (2017) Genomic epidemiology reveals multiple introductions of Zika virus into the United States. Nature 546: 401–405. <u>https://doi.org/ 10.1038/nature22400 PMID: 28538723</u>
- Salje H, Lessler J, Paul KK, Azman AS, Rahman MW, et al. (2016) How social structures, space, and behaviors shape the spread of infectious diseases using chikungunya as a case study. Proceedings of the National Academy of Sciences of the United States of America 113: 13420–13425. <u>https://doi.org/ 10.1073/pnas.1611391113</u> PMID: 27821727
- Vazquez-Prokopec GM, Montgomery BL, Horne P, Clennon JA, Ritchie SA (2017) Combining contact tracing with targeted indoor residual spraying significantly reduces dengue transmission. Science Advances 3.
- Dantés HG, Farfán-Ale JA, Sarti E (2014) Epidemiological Trends of Dengue Disease in Mexico (2000– 2011): A Systematic Literature Search and Analysis. PLoS Neglected Tropical Diseases 8: e3158. https://doi.org/10.1371/journal.pntd.0003158 PMID: 25375162
- 23. Hladish TJ, Pearson CA, Chao DL, Rojas DP, Recchia GL, et al. (2016) Projected Impact of Dengue Vaccination in Yucatan, Mexico. PLoS Negl Trop Dis 10: e0004661. https://doi.org/10.1371/journal. pntd.0004661 PMID: 27227883
- Hernandez-Avila JE, Rodriguez MH, Santos-Luna R, Sanchez-Castaneda V, Roman-Perez S, et al. (2013) Nation-wide, web-based, geographic information system for the integrated surveillance and control of dengue fever in Mexico. PLoS One 8: e70231. <u>https://doi.org/10.1371/journal.pone.0070231</u> PMID: 23936394
- 25. Texier G, Farouh M, Pellegrin L, Jackson ML, Meynard J-B, et al. (2016) Outbreak definition by change point analysis: a tool for public health decision? BMC Medical Informatics and Decision Making 16: 33. https://doi.org/10.1186/s12911-016-0271-x PMID: 26968948
- **26.** Ord JK, Getis A (1995) Local spatial autocorrelation statistics. Distributional issues and an application. Geographical Analysis 27: 286–306.
- 27. Bates D, Mächler M, Bolker B, Walker S (2015) Fitting Linear Mixed-Effects Models Using Ime4. 2015 67: 48.
- Bowman LR, Tejeda GS, Coelho GE, Sulaiman LH, Gill BS, et al. (2016) Alarm Variables for Dengue Outbreaks: A Multi-Centre Study in Asia and Latin America. PLoS ONE 11: e0157971. https://doi.org/ 10.1371/journal.pone.0157971 PMID: 27348752
- Halloran ME, Ferguson NM, Eubank S, Longini IM, Cummings DAT, et al. (2008) Modeling targeted layered containment of an influenza pandemic in the United States. Proceedings of the National Academy of Sciences 105: 4639–4644.

- Bousema T, Griffin JT, Sauerwein RW, Smith DL, Churcher TS, et al. (2012) Hitting Hotspots: Spatial Targeting of Malaria for Control and Elimination. PLOS Medicine 9: e1001165. https://doi.org/10.1371/ journal.pmed.1001165 PMID: 22303287
- Vazquez-Prokopec GM, Spillmann C, Zaidenberg M, Gürtler RE, Kitron U (2012) Spatial Heterogeneity and Risk Maps of Community Infestation by Triatoma infestans in Rural Northwestern Argentina. PLOS Neglected Tropical Diseases 6: e1788. https://doi.org/10.1371/journal.pntd.0001788 PMID: 22905276
- Vanlerberghe V, Gómez-Dantés H, Vazquez-Prokopec G, Alexander N, Manrique-Saide P, et al. (2017) Changing paradigms in Aedes control: considering the spatial heterogeneity of dengue transmission. Rev Panam Salud Publica 41: e16.
- Reiner RC Jr., Achee N, Barrera R, Burkot TR, Chadee DD, et al. (2016) Quantifying the Epidemiological Impact of Vector Control on Dengue. PLOS Neglected Tropical Diseases 10: e0004588. <u>https://doi.org/10.1371/journal.pntd.0004588 PMID: 27227829</u>
- Hladish TJ, Rojas DP, Pearson CB, Gomez-Dantes H, Halloran ME, et al. (in prep) Effectiveness of indoor residual spraying on dengue transmission.
- 35. Manrique-Saide P, Che-Mendoza A, Barrera-Perez M, Guillermo-May G, Herrera-Bojorquez J, et al. (2015) Use of insecticide-treated house screens to reduce infestations of dengue virus vectors, Mexico. Emerg Infect Dis 21: 308–311. https://doi.org/10.3201/eid2102.140533 PMID: 25625483
- 36. Vazquez-Prokopec GM, Lenhart A, Manrique-Saide P (2016) Housing improvement: a novel paradigm for urban vector-borne disease control? Trans R Soc Trop Med Hyg 110: 567–569. <u>https://doi.org/10. 1093/trstmh/trw070 PMID: 27864518</u>
- **37.** Sarti E, L'Azou M, Mercado M, Kuri P, Siqueira JB Jr, et al. (2016) A comparative study on active and passive epidemiological surveillance for dengue in five countries of Latin America. International Journal of Infectious Diseases 44: 44–49. https://doi.org/10.1016/j.ijid.2016.01.015 PMID: 26836763
- LaRocque RC, Breiman RF, Ari MD, Morey RE, Janan FA, et al. (2005) Leptospirosis during Dengue Outbreak, Bangladesh. Emerging Infectious Diseases 11: 766–769. <u>https://doi.org/10.3201/eid1105.041212 PMID: 15890136</u>
- Vazquez-Prokopec G, Montgomery B, Horne P, Clennon J, Ritchie SA (2017) Combining Contact Tracing with Targeted Indoor Residual Spraying Significantly Impacts Dengue Transmission. Science Advances 2.
- 40. Reiskind MH, Lounibos LP (2013) Spatial and temporal patterns of abundance of Aedes aegypti L. (Stegomyia aegypti) and Aedes albopictus (Skuse) [Stegomyia albopictus (Skuse)] in southern Florida. Med Vet Entomol 27: 421–429. https://doi.org/10.1111/mve.12000 PMID: 23278304
- **41.** Stewart-Ibarra AM, Muñoz Á G, Ryan SJ, Ayala EB, Borbor-Cordova MJ, et al. (2014) Spatiotemporal clustering, climate periodicity, and social-ecological risk factors for dengue during an outbreak in Machala, Ecuador, in 2010. BMC Infect Dis 14.
- 42. Vincenti-Gonzalez MF, Grillet ME, Velasco-Salas ZI, Lizarazo EF, Amarista MA, et al. (2017) Spatial Analysis of Dengue Seroprevalence and Modeling of Transmission Risk Factors in a Dengue Hyperendemic City of Venezuela. PLoS Negl Trop Dis 11.
- 43. Ngono AE, Shresta S (2018) Immune Response to Dengue and Zika. Annu Rev Immunol.
- 44. Maslow JN (2017) Vaccine development for emerging virulent infectious diseases. Vaccine.
- Desmond-Hellmann S (2016) Progress lies in precision. Science 353: 731. <u>https://doi.org/10.1126/science.aai7598 PMID: 27540141</u>
- **46.** Dowell SF, Blazes D, Desmond-Hellmann S (2016) Four steps to precision public health Nature 540: 189–191.
- 47. Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, et al. (2013) The global distribution and burden of dengue. Nature 496: 504–507. https://doi.org/10.1038/nature12060 PMID: 23563266
- Messina JP, Kraemer MU, Brady OJ, Pigott DM, Shearer FM, et al. (2016) Mapping global environmental suitability for Zika virus. Elife 5.